Author's response to reviews

Title: Investigating socio-economic-demographic determinants of tobacco use in Rawalpindi, Pakistan

Authors:

Ali Alam (aliyawardalam@gmail.com)
Azhar Iqbal (azhar@heartfile.org)
Khalil Mohamud (drbile@who.org)
Ronld Laporte (ronlaporte@aol.com)
Ashfaq Ahmed (ddgih@hotmail.com)
Sania Nishtar (sania@heartfile.org)

Version: 4 Date: 28 January 2008

Author's response to reviews: see over
Investigating socio-economic-demographic determinants of tobacco use in Rawalpindi, Pakistan

(Revised Cover Letter)

Ali Yawar Alam¹, Azhar Iqbal², Khalif Bile Mohamud³, Ronald E Laporte⁴, Ashfaq Ahmed⁵, Sania Nishtar⁶
Shifa college of Medicine, Islamabad, Pakistan, Heartfile, W.H.O & Ministry of Health Islamabad, Pakistan

1. Assistant Professor, Community Health Sciences, Shifa College of Medicine, Islamabad, Pakistan.
2. Program Officer, Heartfile, Islamabad, Pakistan
3. W.H.O Representative Pakistan
4. Director Disease Monitoring and Telecommunications WHO Collaborating Centre, Professor of Epidemiology, University of Pittsburgh, U.S.A
5. Ministry of Health, Islamabad, Pakistan
6. Founder and President, Heartfile, Islamabad, Pakistan

Why BMC Public Health should publish this Research article:
Tobacco control is an important area of public health worldwide, more so in the developing world where awareness levels regarding their potential for harm are very low. This original Research work conducted in a developing country setting needs a wider audience in other such nations where tobacco consumption and its sequale have wide range health implications. In my view BMC public health is just the right place to publish this manuscript, because of its speedy process, online availability and wide audience.

Corresponding Author

Ali Yawar Alam, MBBS, MS(U.K) MPH(U.S.A)
Assistant Professor, Community Health Sciences, Shifa College of Medicine, Islamabad, Pakistan

Email: aliyawaralam@gmail.com
Telephone: 92-51-4603751
Fax: 92-51-4435046

Ethics Committee Clearance:
Ethical clearance for this study was obtained from ‘Ethics committee at Heartfile, Islamabad, Pakistan,’ in September 2004. Informed consent was obtained from the study participants.
POINT BY POINT RESPONSES

Formatting changes requested

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system. DONE

Title page - This should include the following: the manuscript title (in sentence case i.e. remove all unnecessary capitals), the full names (on one line) and beneath that, associated institutional affiliations for all authors (denoted by superscript numbers). The corresponding author details should then be listed, and a list of all author email addresses following that. The corresponding author should be denoted with the use of a symbol in the author list e.g. *. Also, please align the title page to the left, and ensure that there are no superfluous extras e.g. descriptive words like 'title page¿, 'authors¿, or 'affiliations¿. Word counts, key words, headers/footers, and short/running titles should also be removed. DONE

Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city, country (please remember to include the country). This may mean repeating institutional addresses if the authors are from different departments within an institution. Please do not use full capitals for country names, and please do not include full stops in abbreviated country names. Can we ask you to link each author to his or her corresponding affiliation with the use of superscript numbers. DONE

For example:
Enitan Sawyerr1,2, Emma Veitch3
1 Plant Physiology, Zentrum für Molekularbiologie der Pflanzen (ZMBP), Universität Tübingen, Auf der Morgenstelle1, D-72076 Tübingen, Germany
2 Current address: Protein Interaction laboratory, Dept. Biochemistry and Molecular Biology, University of Southern Denmark, Campusvej 55, 5230 Odense, Denmark
3 Plant Biology, University of Minnesota, 220 Biological Sciences Center, 1445 Gortner Ave., St. Paul, MN55108-1095, USA

Please place the ethics statement within the acknowledgements. Include the funding section within the acknowledgements as well. DONE

Underlining - Please do not underline any section headings. Use bold text to highlight. DONE

Keywords section - Please remove the keywords section from the abstract. DONE

Abstract - Please ensure this is identical in both the manuscript and submission system. DONE

Competing interests - Please write 'The authors declare that they have no competing interests'. DONE

Tables - We are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold. We recommend using the MS Word auto-format template: simple 1. DONE

Figure title - The image file should not include the title (e.g. Figure 1... etc.) or figure number. The legend and title should be part of the manuscript file after the reference list. The figures are numbered automatically in the order in which they are uploaded. DONE

Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: DONE
http://www.biomedcentral.com/info/ifora/figures

Please remove the cover letter from the additional files upload facility. DONE

References - The reference list should contain all authors¿ names, if there are 30 or below, the term
`et al.` should not be used. If there are more than 30 names a minimum of 30 should be listed, and then et al can be used. **DONE**

Please take this opportunity to review your manuscript one last time for any errors or omissions. **DONE**
Submitted manuscripts are presumed to be in their final form when uploaded and modifications are far more difficult to implement post acceptance than they are now.

**Thank You**

**Dr.Ali Yawar Alam**
aliyawaralam@gmail.com

**28th January, 2008**